\begin{table}%t1
\par
\caption{\label{tab:sourcelist}List of observed sources. Columns~2
and~3 give the coordinates used in the observations and for
correlation, Col.~4 gives the source distances in~Mpc. Column~5
gives the classification according to \cite{Veron2001}. }
%\centerline
{
\begin{tabular}{lllccll}
\hline
\hline
Source & 
\multicolumn{1}{l}{\rm RA (J2000)} & 
\multicolumn{1}{l}{\rm Dec (J2000)} & 
$D$ &
Type &
Purpose & 
{\rm Position reference}\\
\hline
\object{NGC~5506} & 14~13~14.87926 & --03~12~27.6514  & 24.7$^{1a)}$ & S1n$^{1b)}$ & Science Target   & Roy et~al. (1997)$^{1c)}$\\
\object{1404-015} & 14~04~45.8949  & --01~30~21.937   &              &             & Phase calibrator & --\\
\object{NGC~7674} & 23~27~56.712   & +08~46~44.14    & 116$^{2a)}$  & S1h$^{2b)}$ & Science Target   & Krichbaum et~al. (1994)$^{2c)}$\\
\object{2327+096} & 23~27~33.5808  & +09~40~09.460   &              &             & Phase calibrator & \cite{1998MNRAS.293..257B}\\
\object{NGC~2110} & 05~52~11.376   & --07~27~22.52    & 31.1$^{3a)}$ & S1i$^{3b)}$ & Science Target   & \cite{1983ApJ...264L...7U}$^{3c)}$\\
\object{0541-056} & 05~41~38.0848  & --05~41~49.395   &              &             & Phase calibrator & --\\
\object{Mrk~1210} & 08~04~05.856   & +05~06~49.83    & 53.9$^{4a)}$ & S1h$^{4b)}$ & Science Target   & \cite{1998ApJ...502..199F}$^{4c)}$\\
\object{0803+043} & 08~03~56.4444  & +04~21~02.724   &              &             & Phase calibrator & \cite{1998MNRAS.293..257B}\\
\hline
\end{tabular}}
\smallskip
Remarks:~$1a$)~\cite{1996ApJS..106...27K};
$1b$)~\cite{2002AaA...391L..21N};
$1c$)~Coordinates from Roy et~al. (1997) observations (unpublished), which based on 
VLA~observations by \cite{1984ApJ...285..439U};
$2a$)~\cite{2000AJ....120.1691N};
$2b$)~\cite{Veron2001}; 
$2c$)~VLA~6~cm (unpublished), 23 25 24.410 +08 30 12.60 (B1950) precessed 
to~J2000 by NRAO's scheduling software SCHED, \cite{1996ApJS..106...27K};
$3a$)~\cite{1995ApJS...99...67N};
$3b$)~\cite{1998csan.book.....V}; 
$3c$)~05~49~46.376  $-$07~28~01.99 (B1950) precessed to J2000 by SCHED;
$4a$)~\cite{1991trcb.book.....D}; 
$4b$)~\cite{1998csan.book.....V}; 
$4c$)~Based on 6~cm VLA~observations from \cite{1998ApJ...502..199F}.
\end{table}